Author's response to reviews

Title: Dynamic navicular motion measured using a stretch sensor is different between walking and running, and between over-ground and treadmill conditions

Authors:

Christian J Barton (christian@completesportscare.com.au)
Simon L Kappel (iha@simonk.dk)
Peter Ahrendt (pah@eng.au.dk)
Ole Simonsen (ohs@rn.dk)
Michael S Rathleff (michaelrathleff@gmail.com)

Version: 5 Date: 2 February 2015

Author's response to reviews:

p values have been italicised as requested.